TCER Working Paper Series

Tokenism in Gender Diversity among Board of Directors

Kan Nakajima Yoko Shirasu Eiji Kodera

March 2024

Working Paper E-201 https://www.tcer.or.jp/wp/pdf/e201.pdf



TOKYO CENTER FOR ECONOMIC RESEARCH 1-7-10-703 Iidabashi, Chiyoda-ku, Tokyo 102-0072, Japan

©2024 by Kan Nakajima, Yoko Shirasu and Eiji Kodera.

All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including ©notice, is given to the source.

Abstract

This study examines the existence of tokenism in Japanese companies after the implementation of corporate governance reforms. We focus on the appointment of female outside directors. The existence of tokenism in corporate boards is an important issue for companies worldwide because it deals with gender diversity in the appointment of board members. Following the Abenomics policy of empowering women, Japan introduced "Japan's Corporate Governance Code" (the Code), which included board reforms such as appointing at least two outside directors. Using a quasi-natural experiment, we examine whether tokenism occurs in Japan, a country with a low female participation level in business. Empirical analysis reveals the occurrence of tokenism at the start of the Code's introduction. Companies appoint two outside director first and a female director later as a token. However, tokenism is not observed when busy female directors with a lot of experience are appointed to the board because they presumably have the expertise and skill.

Kan Nakajima Musashi University Faculty of Economics 1-26-1 Toyotama-kami, Nerima-ku, Tokyo 176-8534, Japan kan.nakajima@cc.musashi.ac.jp Yoko Shirasu TCER and Aoyama Gakuin University Department of Economics 4-4-25 Shibuya, Shibuya-ku, Tokyo 150-8366, Japan shirasu@cc.aoyama.ac.jp

Eiji Kodera Invesco Asset Management (Japan) Ltd. Investment Risk Management & Quantitative Research Roppongi Hills Mori Tower, 6-10-1 Roppongi Minato-ku, Tokyo 106-6114, Japan ekodera@gmail.com

Tokenism in Gender Diversity among Board of Directors

Kan Nakajima

Faculty of Economics, Musashi University, 1-26-1 Toyotama-kami, Nerima-ku, Tokyo 176-8534, Japan, Email: kan.nakajima@cc.musashi.ac.jp

Yoko Shirasu

Department of Economics, Aoyama Gakuin University, 4-4-25 Shibuya, Shibuya-ku, Tokyo 150-8366, Japan, Email: shirasu@cc.aoyama.ac.jp

Eiji Kodera

Invesco Asset Management (Japan) Ltd., Roppongi Hills Mori Tower, 6-10-1 Roppongi Minato-ku, Tokyo 106-6114, Japan, Email: ekodera@gmail.com

Abstract

This study examines the existence of tokenism in Japanese companies after the implementation of corporate governance reforms. We focus on the appointment of female outside directors. The existence of tokenism in corporate boards is an important issue for companies worldwide because it deals with gender diversity in the appointment of board members. Following the Abenomics policy of empowering women, Japan introduced "Japan's Corporate Governance Code" (the Code), which included board reforms such as appointing at least two outside directors. Using a quasi-natural experiment, we examine whether tokenism occurs in Japan, a country with a low female participation level in business. Empirical analysis reveals the occurrence of tokenism at the start of the Code's introduction. Companies appoint two outside directors to meet the formal requirements of the Code. They appoint a male outside director first and a female director later as a token. However, tokenism is not observed when busy female directors with a lot of experience are appointed to the board because they presumably have the expertise and skill.

Keywords: tokenism, in-group bias, female directors, Corporate Governance Code **JEL**: L25; J16; G32; G38

1. Introduction

Tokenism in terms of gender diversity in boardrooms has become a major discussion point globally. Some institutional investors believe that including female directors helps diversify values, foster innovation, and increase shareholder value. Foreign investors have taken the initiative in demanding that companies have diverse boards. Voting criteria now include measures such as opposing the appointment of a CEO if there are no female directors.¹ However, others believe that appointing female on boards is only for "window dressing" or "tokenism" purposes rather than for adding value. Thus, the motivation for appointing female directors remains unclear.

In Japan, the "Womenomics Policy" was launched as part of the Abe Cabinet's Abenomics growth strategy (Ito and Hoshi, 2020). This policy calls for the promotion of women's employment. Female participation in the labor force to solve labor shortages has long been an issue in Japan. The inability of women to return to society after temporarily leaving the labor force for childbirth (M-shaped pattern²) is considered a typical problem of women in the labor force. According to Ito and Hoshi (2020), the M-shaped pattern has improved since 2015 and the number of women in the labor force has increased, leading to the conclusion that the Womenomics policy is successful, at least regarding women's labor force participation. However, the question of female leadership remains even though the policy may have been successful for female workers. The effectiveness of female directors has not been well-studied despite the fact that executive directors are at the top of the corporate hierarchy and are key to effective governance (Gillan, 2006). Japan differs from other developed countries because it has the lowest percentage of female directors in boards among the seven major countries.³ Therefore, determining whether Japanese companies can achieve gender diversity without falling into tokenism during the rapid corporate governance reform is a research topic worth exploring.

As another policy of Abenomics growth strategy, Japan's Corporate Governance Code (the Code) was introduced in 2015. The Code, which is expected to lead to corporate growth and value enhancement, has "Responsibilities of the Board of Directors" as a core principle. This principle requires companies to appoint at least two independent directors.⁴ According to the agency theory, an increase in board

¹ This information is acquired from Nikkei 5/31/2022.

² The M-shaped curve refers to the phenomenon that women drop out of the labor force after childbirth, concentrate on raising their children in the family, and return to the labor force when their children grow up. After their children grow up, they return to the labor force. (Süssmuth-Dyckerhoff et al., 2012)

³ The World Economic Forum's July 2022 Gender Gap Index, which measures the degree of gender equality, ranks Japan 116th out of 146 countries, the lowest in the East Asia and Pacific region, with significantly lower scores in the political and economic areas. Japan ranks 116th out of 146 countries in the East Asia and Pacific region, the lowest in the two areas of political and economic (https://jp.weforum.org/reports/global-gender-gap-report-2022).Western companies lead the way in this area, with the number of companies in the U.S. S&P 500 Index and the U.K. FTSE 350 Composite Index with no female directors close to zero, and women making up 30% of the total number of directors in both cases (Nikkei 2021/11/20).

⁴ The Code is a "soft law" that requires companies to comply with the principles or, if not implemented, to disclose the reasons for not doing so. This differs markedly from Norway and France, where there is strong public sector

independence, as required by the Code, implies diversity in board composition. Moreover, according to the resource dependence theory, female outside directors with different backgrounds from their male counterparts can bring diverse knowledge to management decision-making. Thus, the Code calls for a more diverse boards that are different from Japanese traditional male-dominated board directors in which members are elected based on the "inter group" and "in-group" bias. In other words, advancing board diversity is supported from the perspective of maximizing firm value.

However, several challenges are associated with gender diversity. Kanter (1977) has introduced the concept of "tokenism,"⁵ where substantially outnumbered women are considered symbolic figures who cannot fully exercise their abilities because of the pressures of tokenism. Having two female directors on the board is still considered tokenism because at least three female directors are required.⁶

This study investigates tokenism in the Japanese market. We use the first phase of implementation of the Code in Japan, where female representation is extremely low as a quasi-natural experiment. The results are used to confirm whether tokenism in Japan is the same as that in other countries. Japanese corporate governance is a drastic reform characterized by boards composed mainly of in-group male directors (Aoki et al., 2007). Therefore, we examine the mechanism of tokenism to determine whether the number of female outside directors have increased or whether females have been promoted as corporate directors after the introduction of the Code.

In this study, we collect Japanese board member data from the web and other sources for each company, including information on the number and proportion of board members and the attributes of all female board members. The empirical results show that tokenism occurs when appointing a second outside director; that is, the number of male outside directors increases with the appointment of a first outside director, and a female outside director is subsequently appointed as a token. In Japan, tokenism is seen at the beginning of the Code implementation.

However, tokenism is not observed among busy outside female directors. This may be because the number of busy females outside directors is limited even though they are probably highly qualified.

regulation (Rosenblum, 2019). Note that the 2015 Code did not require board diversification through the appointment of women, and this was included three years later in the revised 2018 Code. Additionally, the legal requirement for the appointment of outside directors has been mandated by the revised Companies Act in 2019.

⁵ The European Institute for Gender Equality (EIGE) defines tokenism as "A policy or practice that is mainly symbolic and involves attempting to meet one's obligations with respect to established targets, such as voluntary or mandated gender quotas, with limited efforts or gestures, especially towards minority groups and women, in a way that does not change male-dominated power and/or organizational arrangements."

⁽ https://eige.europa.eu/publications-resources/thesaurus/terms/1261). This category includes companies that appoint female outside directors to meet the requirements, but have not changed their male-dominated board composition. ⁶ This information is acquired from Nikkei 2021/4/7.

Evidence suggests that the selection of female professionals with relatively high skills has diverse effects on the skills of directors (Hillman et al., 2002).

This study makes several contributions. First, we confirm the tokenism in Japan, which is consistent with previous studies. Second, we find that the driver of Japanese tokenism is not outside but in-country pressure, specifically, the Code. Finally, we reveal that the order of female outside directors' appointment is dependent on the tokenism among outside directors.

The remainder of the study is structured as follows. Section 2 reviews the Japan's Code background focusing on ingroup bias, Section 3 discusses previous studies, while Section 4 establishes our hypotheses. Section 5 presents the methodology, data, and variables. Section 6 presents the empirical results. Lastly, Section 7 concludes the study.

2. Institutional background: The Code and ingroup bias

As pointed in (Aoki et al., 2007), Japanese boards have been composed of male-dominated ingroups (in-group members). In such cases, the boards and its committees are organized in a way that is convenient for men (ingroup bias), where an ingroup bias among male directors forms organizational structures that suit men's interests, such as the appointment of male directors on a rotation basis. Ingroup bias inhibits the following two crucial functions of the board of directors: supervisory and advisory functions (Coles et al., 2008). The board of directors comprises the top ingroup members, such as the CEO, and directors from the ingroups form the hierarchy below. The CEO and board members may have once been superiors and subordinates. Additionally, a director plotting to become the next CEO may be a yes-man to improve the current CEO's impression. Under these circumstances, it is unlikely that a director would express an opposing view to the CEO. Furthermore, because ingroup members are from the same generation, they are colleagues who share the same experiences and values. In such boards, it is not easy to generate diverse opinions and innovations that contribute to improving enterprise value.

To change this situation and enable economic growth, the Tokyo Stock Exchange (TSE) requires that at least one independent outside director or independent outside auditor be appointed in 2009 and at least more than one independent outside director be appointed in 2014. Accordingly, TSE and the Financial Services Agency jointly formulated and published the Code in 2015.

The Code called for a more diverse board, different from a board elected from the ingroup, one with more independence and greater information transparency. Traditionally, even when outside directors are available, they often come from cross-shareholding companies, companies with which the firm conducts business, such as major banks, or the management. The president/CEO has strong authority and a board of male/internal directors and an internal audit department cannot adequately oversee the board. In response to these challenges, the board structure was to be reformed by increasing the number of independent outside directors to provide independent and effective management oversight (Shirai, 2022). Appointing female outside directors with different backgrounds from male directors and from within the company is expected to bring independence to the board and improve its monitoring function. Furthermore, diversity improves the board's advisory function.

The social role theory has been used to explain the gender gap in the boards of directors. It states that stereotypical attitudes emphasizing traditional gender roles act as barriers to the rise of female directors and candidates for director positions. This problem is difficult to overcome. It has been observed to be associated with economic, political, and cultural institutional factors that make change difficult (Chizema et al., 2015). Thus, increasing the number of female directors and women in senior management takes time despite improvements in female labor force participation in Japan.

As described above, we believe that newly appointed female outside directors can improve both the ingroup bias in male-dominated boards and the gender roles that have long persisted in Japan. To overcome these issues, firms should go beyond tokenism and appoint sufficient female outside directors. Accordingly, this study sheds light on tokenism in Japan.

3. Literature review

Kanter (1977) shows how females are treated as tokens in the corporation. Tokenism is the practice of prioritizing diversity as a mere token by including only a small number of females within a maledominant group. This is called tokenism by gender in corporations. Tokenism in the boardroom arises when females are viewed with prejudice and are under mental pressure to conform to the male majority; moreover, they are also unable to show their capabilities and less likely to influence majority groups. Even if they are able to show their abilities, they are considered exceptions instead of the norm. Furthermore, male directors do not like the implicit agreement to include more females on the board. Therefore, they continue to maintain female directors as tokens unless external pressures are introduced. If they add a female member to the board, they are seen as outside (outgroup) members in the male-dominated group (ingroup) to which they belong⁷.

⁷ Social psychology has shown that evaluations of ingroup members by ingroup members tend to be higher than those by outgroup members (Bodenhausen et al., 2012).

By proportion of female employees, organizations are classified into the following four categories: 1) uniform, 2) skewed, 3) tilted, and 4) balanced (Kanter, 1977). 1) The uniform group comprises homogeneous members with identical values in regard to decision-making; 2) the skewed group has a maximum of two heterogeneous individuals, treated as tokens by the dominant group. For example, a board with fewer than two female directors; 3) the tilted composition group is a minority group comprising a certain percentage of the board. The members can have their views reflected as a group within the board, which is why tokenism does not generally occur. A skewed group becomes a tilted group when the proportion of females is 30% or there are three or more females on the board (Konrad et al., 2008). This threshold is called "critical mass"; finally, 4) the balanced group is a minority group that is given a fair chance within the company. A tilted group becomes a balanced group when the proportion of female sis 40% or more.

Many studies have examined the benefits of gender diversity in the boardroom. Generally, the skewed group is at risk of falling into groupthink (Hart, 1991; Matsui, 2020). Moreover, promoting board diversity has been shown to have positive effects. The existence of female board members improves diversity in terms of expertise and skill. Compared to male directors, female directors have higher educational levels and professional backgrounds in law and finance (Hillman et al., 2002). Additionally, they also diversify the skillsets of boards of directors (Kim and Starks, 2016). Specifically, increased diversity in boards, including gender, race, career history, and age diversity, improve the monitoring of inappropriate corporate investment activities, such as overinvestment (Adams and Ferreira, 2009; Harjoto et al., 2018). Guldiken et al. (2019) state that gender diversity on boards contributes to new ideas, avoiding securities fraud, encouraging prudent oversight, and driving ethical corporate behavior.

Knippen et al. (2019) investigate how companies in the U.S. counter external pressure for greater board gender diversity. In the U.S., external pressure for gender diversity on boards has increased the number of companies appointing at least one female director.⁸ Although new directors are usually appointed without increasing the board size, it is found that, in companies with fewer than two female directors, new female directors are likely to be appointed by increasing the board size and not by replacing former male directors. This is because incumbent male directors categorize the new female directors as outgroup members and appoint them by adding new board seats to retain the incumbent male directors. When two or more incumbent female directors already exist, the number of female directors does not increase because they merely pass the number of female directors required. Additionally, they find that new female directors introduced by increasing the board size are less likely to serve on major board committees. Even when female directors are highly qualified, they are less

⁸ Companies with one female director rarely add two or more female directors (Guldiken et al., 2019).

likely to take on chairing key committees (Field et al., 2020). Thus, the effects of board gender diversity are limited. Knippen et al. (2019) consider tokenism as companies' actions with fewer than two female directors in treating them as an outgroup.

However, there is another story about categorizing female directors as an outgroup. For example, Zhu et al. (2014) report that female directors sharing commonalities with incumbent male directors, such as having the same college education, are more likely to hold significant board positions. This is attributed to incumbent male directors reclassifying female directors and accepting them into the group (ingroup). Countries with a higher proportion of females in legislative bodies, at high-ranking and managerial positions, as along as countries with a lower gender income gap, tend to have a higher proportion of female directors, the proportion of female directors is relatively high. For example, companies with nomination committees ensure transparency in selecting and dismissing directors, resulting in a higher proportion of female directors. Moreover, female directors in nomination committees further increase gender diversity in boards (Hutchinson et al., 2015). Guldiken et al. (2019) point out that more female top managers, that is, CEOs, CFOs, senior vice presidents, et cetera, increase female director appointments.

Outside U.S., in France and Italy, companies with family management have more female directors (Nekhili and Gatfaoui, 2013). In Japan, companies with more foreign institutional investors are more likely to appoint females as directors and managers (Kubo and Nguyen, 2021; Mun and Jung, 2018), depending on the firm size and industry (Nguyen and Thai, 2022). Additionally, female director networks promote the appointment of female directors (Raddant and Takahashi, 2022).

Several studies discuss the relationship between the proportion of female directors and financial performance. The evidence of the relationship between female directors and financial and market performance is mixed.⁹ However, Post and Byron (2015) suggest that countries with better investor protection and less gender discrimination tend to have better financial performance (ROA and ROE) when female directors are appointed. This indicates the need to establish social and economic systems to promote the appointment of female directors. In Japanese companies, studies have shown that the appointment of female outside directors increases firm value (Q ratio) (Tanaka, 2019), the appointment of the first female director improves CSR performance (Nguyen and Thai, 2022), and there is a positive relationship between female founder CEOs and Q (Kubo and Nguyen, 2021).

⁹ Some report positive effects on financial and market performance (e.g. Green and Homroy, 2018; Tanaka, 2019). However, some have found no robust positive effects (e.g., Gregory-Smith et al., 2014; Sila et al., 2016) and provide mixed results (e.g. Adams and Ferriera, 2009; Joecks et al., 2013; Post and Byron, 2015).

4. Hypothesis development

Knippen et al. (2019) find that for firms with two or fewer female directors, instead of replacing male directors with female directors, external pressure to correct tokenism increases the number of female board positions and directors. They argue that even if external pressure leads to the promotion of female directors, board diversity remains limited and does not significantly improve. In such cases, the boards are organized to be convenient for males and perpetuate ingroup bias. In the existence of ingroup bias lead by male directors, male directors are appointed on a rotation basis.

Furthermore, female directors are seen as outgroup instead of ingroup members. Therefore, companies do not actively appoint new female directors unless external pressure is applied. However, when there is pressure, current male directors compromise by nominating new female directors and increasing the board size. Thus, because females are treated as outgroup members, their appointments may be accepted because of ingroup bias, but they are not promoted or given key positions. If there is any external pressure for gender diversity on boards, the number of seats on the board is increased to accommodate outgroup members. Therefore, this study focuses on the increase in the number of female outside directors. Knippen et al. (2019) argue that tokenism occurs as an increase in the board's seats when female directors are treated as outgroup members and external pressure is necessary but the results are not substantive.

Kanter (1977) pointed that "the token" has difficulty gaining power even if there are two people in the group. Thus, in this study, we assume that tokenism occurs when the number of female outside directors on the board in period t is one or two as a result of the appointment of female outside directors in period t-1. If the number of outside directors reaches three or more, it is not considered tokenism. This definition is consistent with that of Konrad et al. (2008), who state that three or more members constitutes critical mass. Recently, the 30% threshold has often been used to refer to the critical mass theory. Examples include the 30% Club launched in the UK in 2010 and Japan's policy target.¹⁰ Considering this criterion, the average number of directors in our sample is 8.92. Therefore, 8.92 * 0.3 = 2.676—a level that is close to three directors.

This study investigates whether tokenism occurs in Japan and proposes a hypothesis regarding the mechanism of tokenism for outside board members in the Japanese context. After the introduction of

¹⁰ The Gender Equality Bureau Cabinet Office has been promoting Positive Action, one of the most effective measures to increase women's participation, in order to achieve the goal of "expecting the percentage of women in leadership positions in all sectors of society to be at least 30% by 2020" (June 20, 2003 decision by the Headquarters for the Promotion of Gender Equality, "30% by 2020" goal). In order to achieve this goal, the Gender Equality Bureau is promoting Positive Action, one of the most effective measures to increase women's participation in society.

the Code, because of in-group bias (Aoki et al., 2007), it was likely for tokenism to occur in Japanese companies. This is where a male outside director is appointed first and a second female director is appointed later. If companies already have one female director and more than two outside directors regardless of gender, they do not need to have any additional female directors (It is enough because of the previously appointed female director). Considering that Japanese corporate boards are male-dominated, we expect that an increase in the number of directors leads to the token appointment of female outside directors. Thus, we derive the following hypotheses:

Hypothesis 1: Under tokenism, female outside directors are appointed not to replace male outside directors but to increase the number of directors. In this case, male outside directors are appointed first, followed by female outside directors.

Under tokenism, high-profile female outside directors are expected to be appointed considering the importance of symbolizing tokenism. Therefore, outside directors are expected to be more visible as career-driven female directors. The expectation is that female outside directors who are also outside directors of other companies are more likely to be nominated.¹¹ Most male directors prefer female outside directors who are busy and do not have the time to serve on the board. Thus, we derive the following hypothesis:

Hypothesis 2: Under tokenism, if female outside directors serve on three or more boards (busy outside directors), they are more likely to be appointed.

Token female outside director appointments may be rooted in strong external pressures (Knippen et al., 2019). In Japan, an increase in the proportion of female directors has become a political and social issue. Additionally, foreign investors may plan to target an increase in proportion because of the low proportion of female directors internationally. In recent years, the institutional environment has persuaded companies to organize CSR activities (Campbell, 2007). Based on the institutional theory, companies with global operations are expected to have more opportunities to consider international norms, non-governmental organizations, and global supply chains. Therefore, companies should incorporate gender diversity in their CSR activities. Tokenism may occur when pressure from various stakeholders is strong. Based on this, we derive the following hypothesis:

Hypothesis 3: Under tokenism, the stronger the external pressure, the more the appointment of female outside directors is promoted.

¹¹ Field et al. (2020) find that busy female directors, who concurrently serve on at least three other companies' boards, often have more professional qualifications, extensive experience in other companies, a wide network of directors, and the necessary qualities.

5. Methods and Sample

Research design

First, we discuss the dependent variables. We create a dummy variable for the conditions under which a firm appoints female outside directors. We focus on the following two cases: a) firms appoint one or two female outside directors in period t when they have no female outside director in period t-1 and b) firms appoint two female outside directors in period t when they have one female outside director in period t-1. To test Hypothesis 1, the case in which an increase in the number of outside directors is observed is used as a proxy variable for tokenism (Female1). The case in which there is no increase in the number of outside directors is used as a proxy variable for no tokenism (Female2). If new outside directors are from ingroup members, they are usually appointed to replace their predecessors. However, under tokenism, if new outside directors are from outgroup members, they are considered additions to the board (Knippen et al., 2019). To test Hypothesis 2, we also create new dummy variables where the appointed female outside directors are busy directors who hold three or more concurrent positions in period t. We introduce the following new variables: 1) Female1b is for a "Female1" female director but who is a busy female director; and 2) Female2b is for a "Female2" female director but who is a busy female director.

Furthermore, we confirm that male outside directors are not considered as tokens. For this purpose, we introduce the following new variables: 1) male outside director (Male1) instead of Female1 and 2) male outside director (Male2) instead of Female2. This formulation allows us to benchmark and compare the determinants of standard male outside director appointments.

Next, we discuss the explanatory variables. We introduce several variables related to the appointment of outside directors in period t-1. The ratio of outside directors (R_Odirectors) and number of outside directors (N_Odirectors). We add the following dummy variables: D0_Odirectors takes the value of 1 if there are no outside directors, and 0 otherwise; D1_Odirectors takes the value of 1 if there is one outside director, and 0 otherwise; D2_Odirectors takes the value of 1 if there are two or more outside directors, and 0 otherwise.

To support Hypothesis 1, first, we note there is an increase in the number of the directors (female1=1) in period t. In this study, we analyze the number of female directors from zero to two and more than, because the Code in 2015 requires at least two outside directors. Hence, we consider the following three cases: 1) If there are no outside directors in t-1 (D0 Odirectors), the first outside director is

expected to be appointed a male and not a female in period t.¹² 2) If one male outside director is appointed from the ingroup members (D1_Odirectors) in t-1, second outside director is expected to be female because the female outside director is appointed after male. 3) If there is no increase in the number of the directors (female2=1), 1) and 2), the problems of tokenism do not occur.

Hypothesis 2 is based on Hypothesis 1, but focuses on busy female outside directors. The same pattern is expected for the increase in the number of board members and appointing female outside directors in period t, as mentioned in Hypothesis 1.

Hypothesis 3 considers that external pressure encourages tokenism. As proxies for pressure in the external environment (Hypothesis 3), we use the foreign ownership ratio (Sh_Foreign), float stock ratio (Sh_Float), foreign sales ratio (Fsales), and average number of female outside directors from the same industry (F_Odirectors). From the perspective of maximizing shareholder value, we believe that foreign and domestic institutional investors demand diversity on boards by engaging with the company and exercising their voting rights. Therefore, companies with higher Sh_Foreign and Sh_Float are likely to face greater pressure to promote gender diversity.¹³ Fsales is considered as a proxy for the degree of overseas business development. As companies with overseas operations are more likely to focus on gender diversity as part of their CSR activities, tokenism may occur when pressure from stakeholders, different from shareholders, is greater. F_Odirectors is a proxy for tokenism pressure. A strong sense of alignment with peer firms is expected to exert pressure, resulting in tokenism.

We introduce explanatory variables that indirectly impact tokenism. First, as variables related to the presence of female leadership, we use a dummy variable (D_Zero) that takes the value of 1 if there are zero female officers (directors or auditors), and 0 otherwise. We also introduce a dummy variable (D_Fceo) that takes the value of 1 if the CEO is female, and 0 otherwise. These variables consider the possibility of promoting the appointment of female outside directors as female CEOs or officers. Auditors and directors are included because both are treated as officers under the Japanese corporate law. The number of foreign directors and auditors (Fsize) is related to foreign management. It also considers the possibility of promoting gender diversity. Next, we use the maximum age of the CEO (Maxage) and a multiple representation dummy (D_Multi) as variables representing male-dominated boards. D Multi is a dummy variable that takes the value of 1 if there is more than one representative

¹² This does not necessarily mean an increase in the number of men on the board.

¹³ Considering that the shareholding ratios of domestic institutional investors are not disclosed in the securities statements (Yuka Syoken Hokokusho), proxy variables need to be searched for. Generally, institutional investors do not invest in illiquid stocks. Therefore, because we believe that there is a certain correlation between the shareholding ratio of domestic institutional investors and free float weight, we use the free float weight as a proxy variable. The free float weight data were obtained from Bloomberg.

director, and 0 otherwise. Under the Japanese corporate law, companies are allowed to have more than one representative director. Consequently, some Japanese companies continue to have representation rights even after retiring from honorary positions. Resistance to promoting gender diversity may be stronger in companies with old customs. We also use the variables number of directors (Bsize), average tenure of directors (Btenure), average years of service as a director (Bcareer), and average age of directors (Bage) to control for member type and composition. We use the number of male director turnovers (To_Mdrt) and female director turnovers (To_Fdrt) as variables. The number of director turnovers is considered because they provide opportunities to promote new directors. Theoretically, this should provide opportunities for both men and women to be promoted. However, turnover does not necessarily lead to promotion of female directors, as discussed in Hypothesis1.

We add the logarithm of total asset(Assets), return on assets(ROA), and industry and year dummies as control variables. All explanatory variables use prior-period values to account for endogeneity effects. For Fsales, Assets, and ROA, which are calculated using financial data, we use values winsorized to a threshold of 1% in the cross-section.

We employ a logistic regression analysis defined by the following equation to test the hypotheses: $Logit[y_{i,t} = 1] = \alpha + \beta \cdot x_{i,t-1} + \gamma \cdot Industry_i + \delta \cdot Year_t$

where y represents the dependent variable to which the values Female1, Female2, Female1b, Female2b, Male1, and Male2 are substituted, respectively. x represents the explanatory variables and β represents the regression coefficient. *Industry* is a dummy variable based on the 17-industry classification by the Tokyo Stock Exchange, and γ is the regression coefficient. *Year* is a dummy variable for the fiscal year, and δ is the regression coefficient. Table 1 summarizes this study's variables.

Table1 Variable definitions

Variables	Definition
Female1	Dummy variable that takes 1 if a firm with no female outside directors in period t-1 appoints 1 or 2 female outside directors in period t or if a firm with 1 female outside director in period t appoints 2 female outside directors in period t, with increasing the board of directors, otherwise 0
Female2	Dummy variable that takes 1 if a firm with no female outside directors in period t-1 appoints 1 or 2 female outside directors in period t or if a firm with 1 female outside director in period t-appoints 2 female outside directors in period t, without increasing the board of directors, otherwise 0
Female1b	Dummy variable that takes 1 if a firm with no female outside directors in period t-1 appoints 1 or 2 female outside directors who concurrently hold positions at 3 or more companies in period t of if a firm with 1 outside director in period t-1 appoints 2 female outside directors who concurrently hold positions at 3 or more companies in period t, with increasing the board of
Female2b	directors, otherwise 0 Dummy variable that takes 1 if a firm with no female outside directors in period t-1 appoints 1 or 2 female outside directors who concurrently hold positions at 3 or more companies in period t of if a firm with 1 outside director in period t-1 appoints 2 female outside directors who concurrently hold positions at 3 or more companies in period t, without increasing the board of directors, otherwise 0
Male1	Dummy variable that takes 1 if a firm appoints one or two new male outside directors from
Male2	period t-1 to period t, with increasing the board of directors, otherwise 0 Dummy variable that takes 1 if a firm appoints one or two new male outside directors from period t-1 to period t, without increasing the board of directors, otherwise 0
R_Odirectors	Number of outside directors / total number of directors in period t-1
N_Odirectors	Number of outside directors in period t-1
00_Odirectors	Dummy variable that takes 1 if a firm has no outside directors in period t-1 , otherwise 0
D1_Odirectors	Dummy variable that takes 1 if a firm has 1 outside directors in period t-1 , otherwise 0
D2_Odirectors	Dummy variable that takes 1 if a firm has 2 or more outside directors in period t-1 , otherwise
Sh_Foreign	Number of shares held by foreign investors / number of shares outstanding in period t-1
Sh_Float	The percentage of listed shares that can be traded in the market in period t-1, which is called the Free-Float Weight, is calculated by the Tokyo Stock Exchange
Fsales	Revenues from overseas customers / total revenues in period t-1
F_Odirectors	Average number of female outside directors within each firms' industry classified by 17 Tokyo Stock Exchange industry sectors in period t-1
O_Zero	Dummy variable that takes 1 if a firm has no female board members in period t-1, otherwise 0
D_Fceo	Dummy variable that takes 1 if a firm has female CEO in period t-1, otherwise 0
size	Number of foreign directors in t-1 period
Maxage	Maximum age of the CEO of each firm during the analysis period
D_Multi	Dummy variable that takes 1 if there is more than one director with authority to represent the company in period t-1, otherwise 0
Fo_Mdrt	Number of male directors who left their positions from period t-1 to period t
lo_Fdrt	Number of female directors who left their positions from period t-1 to period t
Bsize	Number of directors in period t-1
Btenure	Average tenure of directors of each firm in period t-1
Bcareer	Average year of careers on the board of directors in period t-1
Bage	Average age of directors in period t-1
Assets	Natural logarithm of total assets in peirod t-1
ROA	EBITDA / total assets in period t-1

Table 1 summarizes the definitions of variables used in the logistic regression analysis.

Sampling and descriptive statistics

A total of 5,847 fiscal year observations of general business companies selected from the Nikkei NEEDS Corporate Basics & Attributes database are analyzed; companies with irregular fiscal year observations are excluded.¹⁴ The sample is limited to companies listed in the First Section of the Tokyo Stock Exchange because the content of the Code required by the exchanges differs by listing category. Additionally, the sample is limited to a "Company with a Board of Company Auditors" because many Japanese firms had a Board of Corporate Auditors during the analysis period. We also considered differences in the institutional design of the board of directors.¹⁵ Furthermore, the sample is limited to firms with audit committees because many Japanese firms have audit committees during the analysis period, in addition to considering the differences in board institutional design. Financial data are obtained from the Nikkei NEEDS Financial Quest database. We use consolidated data for firms publishing consolidated financial statements. The sample period is from 2014 to 2017.

Table 2 presents the descriptive statistics for the sample. Panel A represents the dependent variables and Panel B represents the explanatory variables. The mean values of Female1 and Female2 are 0.015 and 0.009, respectively, indicating that the level of appointment of female outside directors is low. Especially, the mean values for Female1b and Female2b are 0.002 and 0.004, respectively, indicating that the appointment level of busy female outside directors serving on the boards of three or more companies is extremely low. However, the mean values for Male1 and Male2 are 0.077 and 0.028, respectively, which are higher than those for female directors, but still considered low. N_Odirectors in Panel B is 2.11 (25th percentile value is 2), suggesting that companies may have appointed two outside directors during the early part of the analysis period.

¹⁴ The number of non-financial and non-utility firms listed on the First Section of the Tokyo Stock Exchange is 1,737, 1,806, 1,872, and 1,938 for fiscal years 2014-2017, respectively. For companies with Boards of Company Auditors, data are extracted on companies for which director data were reported. Companies with irregular fiscal years were excluded. The number of companies with a Board of a Company Auditors is 1,606 (92.5%), 1,427 (79.0%), 1,404 (75.0%), and 1,410 (72.8%) for year 2014-2017, respectively. According to a survey conducted by the Japan Association of Corporate Directors (2021), among all companies listed on the First Section of the Tokyo Stock Exchange, 91.4%, 78.8%, 75.0%, and 72.7%, in that order from 2014 to 2017, chose a Company with a Board of Company Auditors. This ensures that our sample is representative of the population.

Note that the setting up of various advisory committees in companies with a Board of a Company Auditors around 2015 was only about 10%, according to the TSE Listed Companies White Paper of Corporate Governance 2015. Therefore, it is likely that few companies had set up advisory committees. Therefore, unlike the U.S. example, it is difficult to verify the detailed profiles of outside directors in Japan in 2015, such as knowing whether they held key positions in the board of directors or not.

¹⁵ A possible method of analysis would be to include 'companies with nominating committees' and 'companies with audit and supervisory committee', whose boards of directors consist mainly of outside directors. In fact, the number of these companies is increasing; however, the change in institutional design changes the size of the board of directors because the audit duties previously handled by the statutory auditors are now performed by the directors. As will be discussed later, the change in board size is noise in identifying tokenism, which is why it is necessary to limit the analysis to the sample that did not change its institutional design during the analysis period. In such a case, the sample size would be greatly reduced, and we would not have a sufficient sample of firms that appoint female outside directors, which is the focus of this study. These are also the reasons for limiting our analysis to' company with a board of a company auditors'.

F_Odirectors is 0.21, which is not far from the industry average for a single female outside director. The average value of D_Zero is 0.66, indicating that only one of three companies has female directors, even when we extend the scope to include internal directors and auditors. The average value of D_Fceo is 0.00 (with a maximum value of 1), indicating that few companies have a female CEO. However, the mean value of To_Fdrt is 0.04, indicating that the turnover rate of female directors is low. This is consistent with the low number of female outside directors appointed. The average value of male directors' turnover is 1.68; therefore, one of our concerns is whether female outside directors are appointed to replace male directors.

	Ν	Mean	Sd	Min	P1	P25	P50	P75	P99	Max
			Pane	l A: Depen	dent varia	ables				
Female1	5,847	0.015	0.12	0	0	0	0	0	1	1
Female2	5,847	0.009	0.10	0	0	0	0	0	0	1
Female1b	5,847	0.002	0.04	0	0	0	0	0	0	1
Female2b	5,847	0.004	0.06	0	0	0	0	0	0	1
Male1	5,847	0.077	0.27	0	0	0	0	0	1	1
Male2	5,847	0.028	0.16	0	0	0	0	0	1	1
			Panel	B: Indepen	ndent vari	ables				
$R_Odirectors$	5,847	24.8%	11.3%	0.0%	0.0%	16.7%	23.1%	31.3%	60.0%	83.3%
N_Odirectors	5,847	2.11	0.94	0	0	2	2	2	5	7
D0_Odirectors	5,847	0.02	0.15	0	0	0	0	0	1	1
D1_Odirectors	5,847	0.19	0.40	0	0	0	0	0	1	1
D2_Odirectors	5,847	0.78	0.41	0	0	1	1	1	1	1
Sh_Foreign	5,807	16.2%	12.7%	0.0%	0.5%	5.8%	13.5%	23.7%	54.4%	94.2%
Sh_Float	5,807	52.8%	17.5%	0.0%	0.0%	40.0%	55.0%	65.0%	85.0%	95.0%
Fsales	5,791	18.9%	25.9%	0.0%	0.0%	0.0%	0.0%	35.1%	88.5%	100.0%
$F_Odirectors$	5,847	0.21	0.09	0.07	0.07	0.15	0.20	0.26	0.50	0.50
D_Zero	5,847	0.66	0.48	0	0	0	1	1	1	1
D_Fceo	5,847	0.00	0.06	0	0	0	0	0	0	1
Fsize	5,847	0.09	0.45	0	0	0	0	0	2	7
Maxage	5,847	61.9	7.6	29	41	58	63	67	79	90
D_Multi	5,847	0.55	0.50	0	0	0	1	1	1	1
To_Mdrt	4,437	1.68	2.39	0	0	0	1	2	11	17
To_Fdrt	4,437	0.04	0.21	0	0	0	0	0	1	3
Bsize	5,847	8.92	2.86	3	4	7	9	10	17	30
Btenure	5,847	6.4	3.6	0.5	1.3	3.7	5.5	8.3	17.4	28.6
Bcareer	5,847	19.7	8.5	0.0	2.4	13.3	20.5	26.6	35.5	42.6
Bage	5,847	59.7	4.6	34.8	44.2	57.8	60.6	62.6	68.6	72.2
Assets	5,807	11.5	1.6	7.1	8.1	10.5	11.4	12.5	15.7	17.7
ROA	5,807	9.7%	6.1%	-52.9%	-2.3%	6.2%	8.8%	12.2%	28.1%	88.3%

Table 2 Descriptive statistics

Table 2 shows the descriptive statistics of the variables used in the logistic regression analysis. The sample analyzes consists of general business firms listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a form of company with a board of auditors. The definitions of each variable are summarized in Table 1. For the three variables calculated using financial data, the ratio of overseas sales (Fsales), log of total assets (Assets), and ROA, we use values winorized to a threshold of 1% to reduce the impact of outliers.

Figure 1 shows the number of companies grouped by the number of outside directors as of 2014. A total of 96 companies have no outside directors, 631 companies have one outside director, and 879 companies have two or more outside directors. One possible reason for the large number of companies with one or at least two outside directors in 2014 is that the Tokyo Stock Exchange's listing rules have been revised to improve corporate governance at listed companies. Specifically, effective February 10, 2014, companies have been required to have at least one independent outside director. Before the implementation of the listing rules, firms were only required to have one independent outside director or auditor. While these stock exchange-led reforms are not legally binding, they act as soft laws, as evidenced by the increase in the number of companies that appointed outside directors–six companies, 102 companies, and 1,302 companies from 2014 to 2017, respectively–indicating progress in appointing outside directors. Companies with no outside directors in 2014 appointed one outside directors by 2017. Conversely, more than half of the sample appointed two or more outside directors in 2014. There are two possible cases, one in which these companies further promote their appointment toward the global standard of the majority or two, be satisfied with meeting the Code's requirements.



Figure 1 Number of firms by number of outside directors appointed

Figure 1 is a histogram showing the number of firms by the number of outside directors appointed. Each bar chart represents the number of companies from FY 2014 to FY 2017, starting from the left. The analysis sample comprises general business companies listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of corporate auditors.

Figure 2 shows the number of outside directors by gender. An increase in the number of companies appointing outside directors has increased the number of outside directors. Specifically, male outside directors have increased from 2,225 in 2014 to 2,621 in 2017, while female outside directors have

increased from 203 in 2014 to 315 in 2017.¹⁶ Regarding the rate of increase, these numbers are approximately 1.2 and 1.6 times higher for male and female outside directors, respectively. This indicates that the number of female outside directors has increased, but their share in the total has remained miniscule.





Figure 2 shows the change in the number of outside directors from 2014 to 2017. The figure is a stacked bar chart by gender. The sample analyzes consists of general business companies listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of corporate auditors.

6. Empirical Results

Is the appointment of female outside directors caused by tokenism?

To verify Hypothesis 1, Table 3 shows the logistic regression results, with Female1 and Female2 as the dependent variables. Table 4 presents the logistic regression results for Female1b and Female2b. In Table 4, to test Hypothesis 2, the dependent variable is busy female outside directors who concurrently serve on the boards of three or more other companies. Table 5 shows the logistic regression results with Male1 and Male2 as the dependent variables. The analysis is conducted to test for differences in the determinants of female outsider director appointments. To test Hypothesis 3, the formulation of the explanatory variables used in all estimates is kept the same; however, explanatory variables that do not contribute to the estimate are excluded.¹⁷

Table 3 Regression results on the likelihood of increasing female outside directors

¹⁶ Counting overlapping cases of concurrent service as outside directors of other companies, there were 2,487 male and 236 female outside directors as of 2014 and 2,907 and 372 as of 2017, respectively, confirming little change from the trends in Figure 2.

¹⁷ The dependent variables y=1 in a situation where there are few observations that are the explanatory variables, small variations in the explanatory variables may result in missing regression coefficients that cannot be estimated. Specifically, all or part of D_Fceo, Fsize, D0_Odirectors, and To_Fdrt are missing in Table 4, while part of D_Fceo is missing in Table 5.

			Fema						Fema			
	model 1	model 2	model 3	model 4	model 5	model 6	model 1	model 2	model 3	model 4	model 5	model 6
R_Odirectors		-3.375***						-1.272				
		[-2.87]						[-0.77]				
N_Odirectors			-0.504***						-0.223			
			[-3.12]						[-1.14]			
D0_Odirectors				-0.115						0.88		
				[-0.17]						[1.08]		
D1_Odirectors					1.129***						0.201	
					[4.25]						[0.47]	
D2_Odirectors						-1.214***						-0.36
						[-4.56]						[-0.87
Sh_Foreign	0.0397	0.338	0.415	0.0301	0.265	0.367	2.053*	2.214*	2.248**	2.131*	2.085*	2.134
	[0.03]	[0.25]	[0.32]	[0.02]	[0.21]	[0.28]	[1.79]	[1.94]	[1.98]	[1.84]	[1.83]	[1.87
Sh_Float	-0.382	-0.284	-0.265	-0.381	-0.197	-0.195	1.277	1.322	1.346	1.267	1.324	1.36
	[-0.50]	[-0.38]	[-0.35]	[-0.50]	[-0.26]	[-0.26]	[1.30]	[1.36]	[1.38]	[1.30]	[1.35]	[1.39
Fsales	0.383	0.417	0.404	0.382	0.396	0.406	-1.088	-1.056	-1.058	-1.081	-1.078	-1.06
	[0.63]	[0.66]	[0.64]	[0.63]	[0.62]	[0.64]	[-1.17]	[-1.14]	[-1.14]	[-1.16]	[-1.16]	[-1.15
F_Odirectors	-8.123	-8.195	-8.456	-8.123	-8.118	-8.132	-22.97***	-23.09***	-23.32***	-22.91***	-23.08***	-23.16**
	[-1.07]	[-1.08]	[-1.11]	[-1.07]	[-1.06]	[-1.06]	[-2.74]	[-2.76]	[-2.77]	[-2.73]	[-2.75]	[-2.75
D_Zero	0.434	0.347	0.334	0.435	0.356	0.343	0.569	0.527	0.498	0.564	0.554	0.54
D_hero	[1.54]	[1.20]	[1.13]	[1.55]	[1.21]	[1.16]	[1.64]	[1.53]	[1.46]	[1.62]	[1.61]	[1.56
D_Fceo	2.802***	2.711***	2.697***	2.803***	2.607***	2.614***	2.934*	2.906*	2.902*	2.947**	2.892*	2.865
D_rceo	[3.59]	[3.53]	[3.50]	[3.61]	[3.55]	[3.53]	[1.89]	[1.91]	[1.91]	[2.02]	[1.80]	2.805
Fsize	-0.601	-0.58	-0.573	-0.602	-0.648*	-0.642*	-0.155	-0.146	-0.129	-0.153	-0.165	-0.17
r size	[-1.54]		[-1.49]	[-1.54]		[-1.66]		[-0.35]	[-0.31]	[-0.35]	[-0.38]	[-0.40
M		[-1.51]			[-1.67]		[-0.35]					
Maxage	-0.00249	-0.00514	-0.00518	-0.00233	-0.00147	-0.00301	0.0145	0.0125	0.0117	0.0131	0.0141	0.013
D M I.	[-0.16]	[-0.33]	[-0.34]	[-0.15]	[-0.09]	[-0.19]	[0.63]	[0.53]	[0.50]	[0.56]	[0.61]	[0.57
D_Multi	0.142	0.0998	0.112	0.142	0.117	0.116	0.083	0.0739	0.0803	0.0915	0.0816	0.083
	[0.55]	[0.38]	[0.43]	[0.55]	[0.45]	[0.45]	[0.23]	[0.20]	[0.22]	[0.25]	[0.22]	[0.23
To_Mdrt	-0.0939	-0.104	-0.108	-0.093	-0.106	-0.116	0.331***	0.332***	0.330***	0.328***	0.329***	0.327**
	[-0.93]	[-1.02]	[-1.09]	[-0.93]	[-1.05]	[-1.14]	[4.40]	[4.40]	[4.37]	[4.36]	[4.38]	[4.37
To_Fdrt	-0.251	-0.226	-0.197	-0.252	-0.253	-0.239	0.607	0.612	0.621	0.619	0.608	0.61
	[-0.25]	[-0.23]	[-0.20]	[-0.25]	[-0.25]	[-0.24]	[0.92]	[0.92]	[0.95]	[0.93]	[0.91]	[0.92
Bsize	-0.137**	-0.181***	-0.0989*	-0.138***	-0.104**	-0.0950*	-0.0921	-0.109	-0.0765	-0.0869	-0.0859	-0.0795
	[-2.56]	[-3.23]	[-1.88]	[-2.58]	[-1.96]	[-1.82]	[-1.25]	[-1.46]	[-1.00]	[-1.18]	[-1.14]	[-1.05
Btenure	-0.0149	-0.0272	-0.0284	-0.0141	-0.0282	-0.0358	-0.00222	-0.00678	-0.00973	-0.00629	-0.00473	-0.0080
	[-0.44]	[-0.78]	[-0.81]	[-0.40]	[-0.84]	[-1.06]	[-0.04]	[-0.13]	[-0.19]	[-0.12]	[-0.09]	[-0.16
Bcareer	0.0369*	0.0199	0.0177	0.0372*	0.0218	0.0189	-0.0182	-0.0233	-0.0256	-0.0203	-0.02	-0.022
	[1.87]	[1.04]	[0.92]	[1.91]	[1.14]	[1.03]	[-0.91]	[-1.14]	[-1.28]	[-1.05]	[-0.97]	[-1.11
Bage	-0.0173	0.0102	0.0121	-0.0187	-0.00298	0.00995	0.00913	0.0189	0.0232	0.018	0.012	0.01
	[-0.53]	[0.30]	[0.37]	[-0.54]	[-0.09]	[0.31]	[0.20]	[0.40]	[0.47]	[0.38]	[0.26]	[0.36
Assets	3.577**	3.909***	4.010***	3.588**	4.259***	4.218***	0.91	0.907	0.98	0.865	0.959	0.9
	[2.45]	[2.68]	[2.78]	[2.45]	[2.87]	[2.86]	[0.52]	[0.52]	[0.56]	[0.50]	[0.55]	[0.57
ROA	4.969*	5.191**	5.077*	4.966*	5.270**	5.241**	3.596	3.664	3.665	3.609	3.682	3.75
	[1.93]	[1.98]	[1.93]	[1.93]	[1.99]	[1.98]	[1.18]	[1.21]	[1.20]	[1.22]	[1.17]	[1.22
Constant	-8.206*	-8.802*	-9.514**	-8.166*	-10.99**	-10.34**	1.604	1.805	1.452	1.243	1.332	1.35
	[-1.74]	[-1.84]	[-2.00]	[-1.73]	[-2.31]	[-2.17]	[0.29]	[0.32]	[0.26]	[0.22]	[0.24]	[0.24
Industry FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Ye
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Ye
Observations	3,989	3,989	3,989	3,989	3,989	3,989	3,953	3,953	3,953	3,953	3,953	3,95
Chi-squared	87.29	103.3	107.1	87.74	124.3	122.7	146.6	150.7	149.2	153.2	148.1	150.
p-value for Chi-squared	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
P raise for our squared	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00

Table 3 presents the results of a logistic regression analysis of the determinants of the probability of appointing female outside directors. The sample analysis consists of general business firms listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of corporate auditors. The definitions of each variable are summarized in Table 1. The numbers in the upper row of the table represent regression coefficients, and the numbers in parentheses in the lower row represent z-values based on cluster robust standard errors for each firm. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.

Models 1-6 in each table are formulations to test Hypotheses 1. Model 1 is a baseline model, while Models 2-6 add variations of the appointment of outside directors in the previous year explanatory variable. The numbers in the top row of the table represent the regression coefficients and are preceded by an asterisk to indicate the significance level. The numbers in parentheses in the bottom row represent the z-values. To estimate the standard errors, we use a cluster-robust estimation method for each firm. To test the adequacy of the hypotheses from different perspectives, we interpret the analysis results in Table 3 with reference to the results in Tables 4 and 5.

			Fema	lle1b					Fema	le2b		
	model 1	model 2	model 3	model 4	model 5	model 6	model 1	model 2	model 3	model 4	model 5	model 6
R_Odirectors		0.0716						-1.134				
		[0.03]						[-0.47]				
N_Odirectors			-0.467						-0.227			
			[-1.00]						[-0.68]			
D0_Odirectors										2.351		
										[1.54]		
D1_Odirectors					0.873						-1.018	
					[1.11]						[-0.92]	
D2_Odirectors						-0.525						0.239
						[-0.57]						[0.28]
Sh_Foreign	1.15	1.148	1.528	0.953	1.246	1.293	3.715*	3.834*	3.902*	4.004*	3.768	3.716
	[0.25]	[0.25]	[0.31]	[0.21]	[0.27]	[0.28]	[1.65]	[1.73]	[1.77]	[1.73]	[1.59]	[1.63]
Sh_Float	-1.879	-1.883	-1.748	-2.138*	-1.909	-1.823	0.759	0.861	0.913	0.688	0.575	0.71
	[-1.32]	[-1.35]	[-1.24]	[-1.71]	[-1.43]	[-1.27]	[0.56]	[0.64]	[0.68]	[0.51]	[0.41]	[0.50]
Fsales	0.887	0.889	0.795	1.001	0.879	0.854	-0.993	-0.956	-0.937	-0.981	-0.967	-0.986
	[0.68]	[0.69]	[0.61]	[0.78]	[0.67]	[0.66]	[-1.29]	[-1.26]	[-1.25]	[-1.24]	[-1.25]	[-1.27]
F_Odirectors	14.38	14.41	12.83	17.68	15.49	14.05	8.167	8.108	8.121	8.162	8.074	8.142
	[0.52]	[0.53]	[0.48]	[0.64]	[0.57]	[0.52]	[0.63]	[0.63]	[0.63]	[0.62]	[0.62]	[0.63]
D_Zero	-1.313*	-1.313*	-1.383*	-1.484*	-1.507*	-1.376*	-2.178***	-2.222***	-2.251***	-2.247***	-2.093***	-2.150***
	[-1.76]	[-1.75]	[-1.85]	[-1.76]	[-1.90]	[-1.86]	[-3.61]	[-3.81]	[-3.84]	[-3.74]	[-3.67]	[-3.92]
Maxage	0.00578	0.00573	0.00898	-0.00058	0.00533	0.00766	0.0288	0.027	0.0258	0.0279	0.0303	0.0293
	[0.11]	[0.11]	[0.17]	[-0.01]	[0.10]	[0.14]	[1.02]	[0.89]	[0.84]	[0.98]	[1.07]	[1.03]
D_Multi	0.39	0.39	0.374	0.356	0.41	0.401	1.09	1.077	1.073	1.104	1.1	1.093
	[0.41]	[0.41]	[0.39]	[0.38]	[0.45]	[0.42]	[1.59]	[1.58]	[1.56]	[1.59]	[1.62]	[1.61]
To_Mdrt	-0.0108	-0.0107	-0.0275	0.00283	-0.0334	-0.0261	0.328**	0.326**	0.322**	0.328**	0.336**	0.330**
	[-0.04]	[-0.04]	[-0.10]	[0.01]	[-0.11]	[-0.09]	[2.16]	[2.17]	[2.17]	[2.14]	[2.21]	[2.21]
To_Fdrt							0.47	0.451	0.444	0.479	0.454	0.465
							[0.81]	[0.78]	[0.77]	[0.81]	[0.78]	[0.79]
Bsize	-0.0652	-0.0641	-0.026	-0.0783	-0.0414	-0.0473	-0.345***	-0.369***	-0.329***	-0.336***	-0.360***	-0.350***
	[-0.35]	[-0.34]	[-0.13]	[-0.42]	[-0.21]	[-0.24]	[-3.49]	[-3.46]	[-3.19]	[-3.33]	[-3.70]	[-3.52]
Btenure	0.0875	0.088	0.0685	0.117	0.084	0.0779	-0.0984	-0.103	-0.107	-0.115	-0.0921	-0.0958
	[1.19]	[1.08]	[0.81]	[1.31]	[1.07]	[1.04]	[-1.23]	[-1.16]	[-1.16]	[-1.33]	[-1.12]	[-1.15]
Bcareer	-0.0479	-0.0476	-0.0548*	-0.0487	-0.0560*	-0.0522	-0.0106	-0.0143	-0.0164	-0.0123	-0.00637	-0.00933
	[-1.36]	[-1.45]	[-1.82]	[-1.35]	[-1.69]	[-1.61]	[-0.31]	[-0.42]	[-0.49]	[-0.36]	[-0.18]	[-0.27]
Bage	0.041	0.0407	0.0595	0.0514	0.0531	0.0473	0.00575	0.014	0.0192	0.0315	-0.00105	0.00266
	[0.34]	[0.35]	[0.49]	[0.39]	[0.45]	[0.41]	[0.10]	[0.22]	[0.29]	[0.48]	[-0.02]	[0.04]
Assets	2.776	2.765	3.121	2.55	3.009	2.905	0.608	0.646	0.698	0.4	0.411	0.557
	[0.49]	[0.48]	[0.55]	[0.45]	[0.52]	[0.51]	[0.27]	[0.29]	[0.31]	[0.17]	[0.18]	[0.25]
ROA	-4.353	-4.368	-4.025	-5.613	-4.87	-4.321	-2.62	-2.562	-2.661	-3.037	-2.898	-2.671
	[-0.80]	[-0.84]	[-0.69]	[-1.19]	[-0.85]	[-0.77]	[-0.54]	[-0.53]	[-0.55]	[-0.61]	[-0.56]	[-0.54]
Constant	-33.70*	-33.34	-34.87*	-34.18*	-35.08*	-33.55*	-8.532	-8.431	-8.886	-9.446	-7.558	-8.446
	[-1.65]	[-1.64]	[-1.83]	[-1.65]	[-1.72]	[-1.68]	[-1.13]	[•1.11]	[-1.18]	[-1.22]	[-1.01]	[-1.12]
Industry FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	2,102	2,102	2,102	2,054	2,102	2,102	3,038	3,038	3,038	3,038	3,038	3,038
Chi-squared	NA	939.7	NA	806.1	872	911.2	198	204.5	200.1	182.7	187	199.1
p-value for Chi-squared	1.11	0.00	1.11	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00

Table 4 Regression results on	the likelihood of ind	reasing the number	er of busy fema	ale outside directors

Table 4 presents the results of a logistic regression analysis of the determinants of the probability of appointing a busy female outside director who concurrently serves on at least three other company boards. The sample analyzed consists of general business companies listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of corporate auditors. The definitions of each variable are summarized in Table 1. The numbers in the upper row of the table represent regression coefficients and those in parentheses in the lower row represent z-values based on cluster robust standard errors for each firm. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.

			Ma	le1					Mal	e2		
	model 1	model 2	model 3	model 4	model 5	model 6	model 1	model 2	model 3	model 4	model 5	model 6
R_Odirectors		-10.33***						-6.324***				
		[-10.21]						[-5.72]				
N_Odirectors			-1.452***						-0.831***			
			[-10.62]						[-5.57]			
D0_Odirectors				1.617^{***}						0.192		
				[5.77]						[0.37]		
D1_Odirectors					1.705***						1.551***	
					[11.97]						[7.33]	
D2_Odirectors						-2.352***						-1.672***
						[-15.47]						[-7.84]
Sh_Foreign	0.486	1.621**	1.688**	0.675	1.087*	1.575**	0.106	0.672	0.601	0.123	0.431	0.573
	[0.76]	[2.40]	[2.50]	[1.05]	[1.69]	[2.37]	[0.11]	[0.66]	[0.59]	[0.13]	[0.43]	[0.57]
Sh_Float	-1.400***	-1.210***	-1.206***	-1.455***	-1.197***	-1.148***	0.783	1.112*	1.136*	0.783	1.154*	1.177**
	[-3.79]	[-3.06]	[-3.05]	[-3.93]	[-3.07]	[-2.86]	[1.36]	[1.83]	[1.86]	[1.36]	[1.94]	[1.97]
Fsales	-0.196	-0.208	-0.153	0.245	-0.131	-0.134	-0.108	-0.0971	-0.117	-0.112	-0.0286	-0.0536
	[-0.75]	[-0.74]	[-0.55]	[-0.94]	[-0.48]	[-0.47]	[-0.27]	[-0.23]	[-0.28]	[-0.28]	[-0.07]	[-0.13]
F_Odirectors	-1.581	-1.906	-2.333	-1.473	-1.778	-1.998	9.671*	9.369*	9.113*	9.691*	9.189*	9.354*
1_Ouncetors	[-0.45]	[-0.53]	[-0.65]	[-0.41]	[-0.50]	[-0.56]	[1.86]	[1.82]	[1.77]	[1.87]	[1.80]	[1.83]
D_Zero	0.309**	0.123	0.0745	0.313**	0.131	0.0705	0.217	0.0678	0.0185	0.216	0.0546	0.0406
D_Zelo	[2.44]	[0.88]	[0.55]	[2.42]	[0.98]	[0.51]	[1.13]	[0.34]	[0.09]	[1.13]	[0.27]	[0.20]
D Essa							[1.10]	[0.34]	[0.09]	[1.13]	[0.27]	[0.20]
D_Fceo	0.547	0.388	0.342	0.538	0.374	0.346						
п.	[1.32]	[0.65]	[0.62]	[1.16]	[0.59]	[0.52]	0.040*	0.00	0.050	0.040*	0.104	0.100
Fsize	-0.369**	-0.443**	-0.463**	-0.357**	-0.517***	-0.584***	0.248*	0.235	0.256	0.248*	0.184	0.183
	[-2.07]	[-2.47]	[-2.50]	[-2.04]	[-2.89]	[-3.12]	[1.69]	[1.55]	[1.63]	[1.69]	[1.24]	[1.24]
Maxage	0.0135	0.00839	0.00677	0.0125	0.0126	0.0113	0.0134	0.00684	0.00626	0.0133	0.00952	0.00855
	[1.60]	[0.95]	[0.79]	[1.48]	[1.39]	[1.22]	[0.93]	[0.46]	[0.43]	[0.92]	[0.65]	[0.58]
D_Multi	0.125	0.0795	0.11	0.116	0.134	0.127	-0.18	-0.224	-0.193	-0.18	-0.174	-0.175
	[1.07]	[0.64]	[0.90]	[1.00]	[1.08]	[0.99]	[-0.95]	[-1.17]	[-1.02]	[-0.95]	[-0.91]	[-0.92]
To_Mdrt	-0.248***	-0.298***	-0.304***	-0.271***	-0.277***	-0.313***	0.202***	0.199^{***}	0.190***	0.201***	0.191***	0.181***
	[-4.17]	[-4.83]	[-5.09]	[-4.57]	[-4.57]	[-5.18]	[3.93]	[3.85]	[3.74]	[3.88]	[3.86]	[3.71]
To_Fdrt	-0.468	-0.365	-0.326	-0.413	-0.44	-0.374	1.156^{***}	1.245^{***}	1.261^{***}	1.160***	1.249^{***}	1.289***
	[-0.81]	[-0.62]	[-0.57]	[-0.72]	[-0.76]	[-0.64]	[3.13]	[3.34]	[3.21]	[3.14]	[3.41]	[3.56]
Bsize	-0.193***	-0.330***	-0.107***	-0.181***	-0.139***	-0.111***	-0.0378	-0.106**	0.0117	-0.0367	0.0106	0.0194
	[-6.80]	[-10.37]	[-3.89]	[-6.44]	[-4.91]	[-3.97]	[-0.92]	[-2.36]	[0.29]	[-0.89]	[0.27]	[0.49]
Btenure	-0.00538	-0.0454^{**}	-0.0471***	-0.0192	-0.024	-0.0456**	-0.00403	-0.0272	-0.0291	-0.00507	-0.0251	-0.0315
	[-0.34]	[-2.35]	[-2.60]	[-1.14]	[-1.34]	[-2.53]	[-0.16]	[-1.07]	[-1.15]	[-0.19]	[-0.98]	[-1.24]
Bcareer	0.0408***	0.000279	-0.000504	0.0367***	0.0149	0.00253	0.0188	-0.00528	-0.00688	0.0184	-0.00213	-0.00546
	[4.30]	[0.03]	[-0.06]	[3.92]	[1.54]	[0.27]	[1.49]	[-0.45]	[-0.58]	[1.48]	[-0.18]	[-0.47]
Bage	-0.0661***	0.00276	0.00632	-0.0494***	-0.0366**	-0.00688	-0.0312	0.0161	0.0192	-0.0295	-0.00182	0.0115
	[-4.05]	[0.15]	[0.36]	[-2.93]	[-2.06]	[-0.39]	[-1.19]	[0.56]	[0.68]	[-1.09]	[-0.07]	[0.42]
Assets	1.427**	2.265***	2.361***	1.491**	2.213***	2.565***	0.13	0.432	0.703	0.123	0.719	0.727
	[2.05]	[2.84]	[3.06]	[2.08]	[3.00]	[3.29]	[0.12]	[0.39]	[0.63]	[0.12]	[0.64]	[0.65]
ROA	-1.453	-0.99	-1.214	-1.468	-0.985	-1.02	0.454	0.948	0.985	0.457	1.211	1.188
	[-1.13]	[-0.70]	[-0.90]	[-1.12]	[-0.73]	[-0.76]	[0.23]	[0.45]	[0.47]	[0.23]	[0.58]	[0.57]
Constant	-1.086	-2.102	-3.795	-2.131	-4.927**	-5.124**	-7.954**	-8.369**	-9.861***	-8.040**	-11.10***	-10.29***
	[-0.49]	[-0.87]	[-1.61]	[-0.95]	[-2.10]	[-2.16]	[-2.42]	[-2.50]	[-2.93]	[-2.44]	[-3.28]	[-3.06]
Industry FE	[-0.49] Yes	[-0.87] Yes	Yes	[-0.95] Yes	Yes	[-2.16] Yes	Yes	[-2.50] Yes	[-2.95] Yes	Yes	[-5.26] Yes	Yes
Year FE	Yes	Yes		Yes	Yes	Yes			Yes	Yes	Yes	Yes
			Yes				Yes	Yes				
Observations	3,989	3,989	3,989	3,989	3,989	3,989	3,941	3,941	3,941	3,941	3,941	3,941
Chi-squared	272.9	329	323.6	299.4	469.6	535.5	91.83	125.2	121	92.04	148.7	155.5
p-value for Chi-squared	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
R-squared	0.118	0.195	0.202	0.131	0.182	0.223	0.046	0.075	0.079	0.047	0.090	0.094

Table 5 Regression results on the likelihood of increasing the number of male outside directors

Table 5 presents the results of a logistic regression analysis of the determinants of the probability of appointing male outside directors. The sample analyzed consists of general business firms listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of corporate auditors. The definitions of each variable are summarized in Table 1. The numbers in the upper row of the table represent regression coefficients and those in parentheses represent z-values based on cluster robust standard errors for each firm. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.

First, we examine the impact of the appointment of outside directors in the previous year. Given that it is difficult to directly interpret the regression coefficient values of the logit estimation, we only check the sign of the coefficients. Below, we discuss the economic significance of the regression coefficients. In the results for Female1 in Table 3, R Odirectors, N Odirectors, and D2 Odirectors are significantly negative, whereas D1 Odirectors is significantly positive. Conversely, no significant trend is observed for Female2. These results suggest that the lower ratio of outside directors and smaller the number of outside directors, the more likely their companies are to appoint female outside directors to increase the number of board members. However, the results are insignificant when boards have no outside directors in the previous year, and the appointment probability declines when boards have two outside directors. Thus, the aforementioned trend is limited to cases in which boards appoint one outside director and then appoint a female as the second outside director. The results for Male1 in Table 5 show that D0 Odirectors is significantly positive, suggesting that males tend to be appointed first as outside directors. D1 Odirectors is also significantly positive, indicating that females are not the only ones appointed as second outside directors. Table 3 shows that the coefficients for R Odirectors, N Odirectors, and D2 Odirectors are significantly negative, indicating that companies are satisfied only fulfilling the external standards required by the Code. This result is consistent with that of Farrell and Hersch (2005), who have found that the likelihood of appointing an additional female director significantly decreases when the board already has one female director. Therefore, the increase in the number of outside directors in the previous year, along with the increase in the number of directors in the current year, is a female-specific trend (Hypothesis 1). Conversely, Table 4 shows no significant trend for either Female1b or Female2b. Therefore, our results do not support Hypothesis 2.

Next, we review the impact of external environmental pressures in Table 3. We examine Sh_Foreign, Sh_Float, Fsales, and F_Odirectors. Sh_Foreign is significantly positive for Female2; almost all are significant at the 10% level. F_Odirectors is significantly negative, while Sh_Foreign is positive at the 10% level, but the consistency declines. The disciplining effect of foreign investors on Japanese firms pointed out in previous studies may promote essential gender diversity efforts instead of tokenism. However, the results in Table 5 indicate that, while no significant trend is observed for Male2, a there is a significantly positive trend for Male 1, but this trend is not conspicuous. Thus, the mechanism by which foreign investors promote the appointment of outside directors tends to be non-uniform across genders. Notably, Sh_Float, which does not affect women, is significantly negative for Male1. Meanwhile, Sh_Float is positive for Male2; although, the trend is inconsistent. Therefore, domestic institutional investors, aware of the difficulty of improving gender diversity in Japan, could have urgently promoted the appointment of male outside directors. The sign of F_Odirectors differs from our expectation. This may be because firms with no female outside directors have promoted

appointments, whereas firms that appointed one female outside director captured the trend without promoting appointments. This corporate behavior may have synchronously occurred within the industry. Fsales is insignificant in all cases, indicating that foreign sales do not place pressure on overseas business activities. Given these results, we conclude that the external pressure presented in Hypothesis 3 does not induce tokenism. Instead, some pressure promotes the appointment of female outside directors in a way that does not increase the number of directors.

The two significant variables having indirect effects on tokenism are D_Fceo and Bsize. The sign of D_Fceo indicates that female CEOs do not promote tokenism. However, given that the results in Table 3 show a consistent trend even for Female2, albeit at a lower level of significance, the results may indicate that companies promote women while taking care of its male-dominated board composition by increasing the number of board members as needed. Bsize is significantly negative, indicating that the larger the board of directors, the less likely it is to appoint women to by increasing the number of directors. However, in the case of the appointment of busy female outside directors in Table 4, Bsize is insignificant for Female1b, but the same trend is observed for Female2b. Additionally, Maxage, D Multi, Btenure, Bcareer, and Bage are all insignificant, contrary to expectations.

Interestingly, the sign of the coefficient differs for appointing outside directors after a director leaves the board differ. Director turnover can trigger the appointment of female outside directors. However, only in the cases of Female2 in Table 3 and Female2b in Table 4 is To_Mdrt significant. Thus, director turnover does not induce tokenism, suggesting that tokenism may be a distinctly different mechanism from the usual election and dismissal processes that do not involve an increase in the number of directors. Another important point is the difference in the significance of the number of departures between Female2 and Male2. Table 3 shows that To_Mdrt is significantly positive, whereas that To_Fdrt is insignificant. However, Table 5 shows that both To_Mdrt and To_Fdrt are significantly positive. Although various factors may be responsible for the turnover of female directors not leading to the appointment of new female outside directors, this may indicate a lack in progress in terms of gender diversity. Finally, Assets and ROA are both significantly positive for Female1. Given that both variables are insignificant for Female2 and Male2, tokenism tends to occur in situations where firm performance is good.

The analysis results suggest that Japanese firms first appoint one male outside director and subsequently appoint a female director as tokenism (Hypotheses 1). Additionally, the results show that firms already with two or more outside directors, regardless of gender, do not appoint more outside directors. However, because no consistent trend is found regarding the tendency to appoint busy female outside directors, Hypothesis 2 is rejected. Busy female outside directors are expected to have

more experience and skills in board service. Therefore, they are not suitable for token appointments in Japan. Regarding external pressures on firms, unlike Knippen et al. (2019), we do not find the tendency to promote tokenism (Hypothesis 3). Instead, our results suggest that some of these pressures may improve the function of the board of directors.

Marginal effects on the likelihood of increasing the number of outside directors

We examine the economic significance of outside director appointments. Specifically, we estimate the marginal effect of previous outside director appointment status on the probability of appointing another outside director. Considering that this analytical model is nonlinear, we estimate the marginal effect for the average firm (around the mean). Table 6 presents the marginal effect and the numbers in parentheses represent the significance level. The dependent variables used in Tables 3 and 5 are horizontally arranged and the proxy variables for the appointment of outside directors in period t-1 used in Models 2–6 are vertically arranged.

	Female1	Female2	Male1	Male2
R_Odirectors (model 2)	-0.048***	-0.011	-0.613***	-0.186***
	[-2.69]	[-0.77]	[-12.60]	[-6.36]
N_Odirectors (model 3)	-0.007***	-0.002	-0.079***	-0.024***
	[-2.99]	[-1.14]	[-14.68]	[-6.45]
D0_Odirectors (model 4)	-0.002	0.007	0.119***	0.006
	[-0.17]	[1.08]	[5.52]	[0.37]
D1_Odirectors (model 5)	0.015***	0.002	0.112***	0.044***
	[3.94]	[0.47]	[11.55]	[7.70]
D2_Odirectors (model 6)	-0.016***	-0.003	-0.139***	-0.047***
	[-4.21]	[-0.88]	[-14.73]	[-8.26]

Table 6 Marginal effects on the likelihood of increasing the number of outside directors

Table 6 shows the results of estimating the effect of the previous year's outside director status on the probability of appointing an outside director for the year in question. Specifically, we estimate the marginal effects using the mean of each variable. The sample analyzed is general business firms listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of auditors. The definitions of each variable are summarized in Table 1. The numbers in the upper row of the table represent regression coefficients and those in parentheses represent z-values based on cluster robust standard errors for each firm. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.

The analysis indicates that when the number of outside directors appointed in the previous year is zero, the probability of the appointment of Female1 remains the same, while that for Male1 increases by 11.9%. When there is one outside director, the probability of appointing Female 1 increases by 1.5%, while that of appointing Male 1 increases by 11.2%. However, when there are already two or more outside directors, the probability of appointing Female1 decreases by 1.6%, while that of Male1 decreases by 13.9%. These results show a difference of more than ten percentage points in the

probability of appointment, indicating that the impact on the appointment of female outside directors is small. The same is true for the ratio and number of outside directors, suggesting that one reason for low economic significance is the lack of progress in appointing outside directors after meeting the external standards of the Code. The results show that, although tokenism is observed during the appointment of female outside directors, its economic impact is limited.

Prioritizing the appointment of female outside directors

To confirm the robustness of the results for Hypothesis 1, we create dummy variables for the appointment of outside directors in the previous year, separately for men and women. For example, D1_Odirectors for Female is a dummy variable that takes the value of 1 if there is one female outside director in period t-1, and 0 otherwise. Other formulations are the same as those listed in Tables 3 and 5.

	Female1	Female2	Male1	Male2
D1_Odirectors (model 5)	1.129***	0.201	1.705***	1.551***
	[4.25]	[0.47]	[11.97]	[7.33]
D2_Odirectors (model 6)	-1.214***	-0.366	-2.352***	-1.672***
	[-4.56]	[-0.87]	[-15.47]	[-7.84]
D1_Odirectors for Male	0.526**	-0.171	1.140***	0.963***
	[2.17]	[-0.45]	[8.86]	[4.83]
D2_Odirectors for Male	-0.694***	0.0383	-1.914***	-1.288***
	[-2.83]	[0.10]	[-13.31]	[-6.19]
D1_Odirectors for Female	-1.122**	-1.140**	-0.249	-0.0503
	[-2.30]	[-1.99]	[-1.15]	[-0.16]
D2_Odirectors for Female	NA	NA	-1.06	0.42
			[-1.04]	[0.73]

Table 7 Impacts of the appointment of outside directors in the previous year by gender

Table 7 shows the estimated impact of the previous year's status of outside directors by gender on the appointment of outside directors in the relevant fiscal year. The sample analyzed consists of general business companies listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of auditors. The definitions of each variable are summarized in Table 1. The numbers in the upper row of the table represent regression coefficients and those in parentheses represent z-values based on cluster robust standard errors for each firm. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.

The analysis results show that the signs of D1_Odirectors for Female1 and Female2 are significantly negative. Thus, the probability of appointing a new female outside director is significantly lower when there is only one female outside director, supporting the existence of tokenism. Conversely, the signs of D1_Odirectors for Male1 and Male2 are significantly positive. Additionally, the sign for Female1 is also significantly positive, confirming that the appointment of a female as the second outside director is accompanied by an increase in the number of directors.

Next, we examine whether the appointment of females as internal directors and auditors is a form of tokenism.¹⁸ Gender diversity on boards is promoted to ensure that the diverse skillsets of outside directors are reflected in decision-making. However, tokenism may arise when a company appoints a female internal director or auditor instead of a female outside director. Therefore, we create a dummy variable (D1_Female_Auditors) that takes the value of 1 when there is one female auditor in period t-1, and 0 otherwise, and another dummy variable for female internal directors (D1_Female Internal directors). We rerun the test under the same formulation as before.

Table 8 Impacts of the appointment of female internal directors and auditors in the previous year

	Female1	Female2	Male1	Male2
D1_Female_Auditors	1.102**	0.776	-0.176	-0.213
	[2.39]	[1.35]	[-0.82]	[-0.66]
D1_Female Internal directors	-0.495	0.498	0.478*	0.0755
	[-0.82]	[0.69]	[1.83]	[0.17]

Table 8 shows the estimated impact on the appointment of outside directors for cases with one female statutory auditor and one female in-house director in the previous fiscal year. The sample analyzed consists of general business companies listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a board of corporate auditors. The definitions of each variable are summarized in Table 1. The numbers in the upper row of the table represent regression coefficients and those in parentheses represent z-values based on cluster robust standard errors for each firm. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.

Table 8 shows that most signs of the coefficients are insignificant and that all variables are positive, suggesting that this possibility is unlikely. The significantly positive sign of D1_Female_Auditors for Female1 suggests that firms with female auditors tend to appoint female outside directors as tokens. Considering that the study period coincides with the period when Japanese companies started to appoint female outside directors, many companies in the male-dominated Japanese society could have had a negative perception about appointing female outside directors, even if they are "token" female directors. In such a situation, the challenge of appointing female outside directors as tokens may be lowered because of less resistance to women in companies with women as auditors.¹⁹ The sign of D1_Female Internal directors is positive for Male1 at the 10% level, indicating that having one woman on the board of directors promotes the appointment of male outside directors. This may be because

¹⁸ Kirsch and Wrohlich (2020) that when the German law requires a 30% gender quota for the supervisory board (Gesetz für die gleichberechtigte Teilhabe von Frauen und Männern an Führungspositionen), the gender quota for supervisory boards (supervisory boards) tends to positively impact the presence of women on the executive board (executive boards).

¹⁹ Another possible interpretation is the possibility of appointing a female outside statutory auditor as an outside director to satisfy the formal requirements. In practice, the case of appointing a previously acquainted auditor who is convenient for management as a director (and vice versa) without considering the existence of their skills and suitability as a director is well-known as a lateral appointment. It has been pointed out that this behavior may hamper corporate governance. A review of this sample shows that the number of cases in which a female outside corporate auditor from the previous year was appointed as an outside director in the current year was limited to five observations, limiting the possibility that a lateral appointment may have occurred.

companies consider that one female inside director on the board of directors is sufficient to support the appointment of male outside directors. Although this trend indirectly affects the appointment of female outside directors, it can be considered as evidence of tokenism.

Does the appointment of female outside directors through tokenism improve a firm's performance?

We do not expect tokenism to have any effect on corporate performance because it does not improve the function of the board of directors. Therefore, we examine corporate performance changes before and after female outside directors' appointments. Specifically, we conduct propensity score matching (PSM) under the formulation of Model 3 and analyze the differences in ROA, Tobin's q (Q), and abnormal accruals (ABACC) as proposed by Dechow et al. (1995). We ascertain tokenism's impact on audit fee (Audit Fee).²⁰ ROA and Q are proxy variables for the advisory function, while ABACC and audit fee are proxy variables for the monitoring function. The PSM method sets the caliper in the empirically used range of 0.2σ . The PSM method is one-to-one matching without replacement. To confirm the validity of the parallel trend assumption, we also check the increments from period t-2 to period t and from period t to period t+2.

²⁰ Univariate comparisons of the outcome variables before and after matching are shown in Appendix 1.

			Fem	ale1			Male1				
		Treat	Control	Difference	t-stat	Treat	Control	Difference	t-stat		
ΔROA	[t-2,t]	0.0%	1.0%	-1.0%	-1.02	-0.3%	0.1%	-0.4%	-0.97		
		35	35			226	227				
	[t-1,t+1]	0.0%	-0.2%	0.1%	0.14	-0.1%	-0.1%	0.1%	0.12		
		34	35			225	220				
	[t,t+2]	-1.5%	-1.8%	0.4%	0.28	-0.3%	-0.7%	0.4%	0.91		
		33	35			223	216				
$\Delta \mathbf{Q}$	[t-2,t]	0.097	0.520	-0.423	-1.91 *	0.008	0.063	-0.056	-1.05		
		35	34			224	225				
	[t-1,t+1]	0.013	0.107	-0.095	-0.41	-0.018	0.065	-0.084	-1.19		
		34	35			225	219				
	[t,t+2]	0.014	-0.089	0.103	0.57	0.058	0.062	-0.003	-0.05		
		33	35			223	214				
AABACC	[t-2,t]	-0.003	-0.005	0.002	0.21	0.000	-0.001	0.001	0.20		
		32	32			205	204				
	[t-1,t+1]	-0.011	-0.007	-0.004	-0.55	-0.002	-0.001	-0.001	-0.26		
		32	32			204	198				
	[t,t+2]	0.001	-0.008	0.009	1.05	-0.002	-0.001	-0.001	-0.26		
		29	32			201	189				
ΔAudit Fee	[t-2,t]	5.5	5.8	-0.4	-0.05	4.0	2.2	1.8	0.48		
		35	35			215	205				
	[t-1,t+1]	1.1	9.3	-8.2	-1.16	1.4	3.6	-2.2	-0.96		
		33	35			217	203				
	[t,t+2]	3.9	10.7	-6.8	-0.78	4.6	4.8	-0.2	-0.06		
		32	35			213	202				

Table 9 Difference-in-difference analysis

Table 9 presents the results of the difference-in-differences analysis for the periods before and after the appointment of outside directors. The numbers in the table represent the mean value of the change in the outcome variable in each period. The numbers below the mean represent the number of observations. The sample analyzed comprises general business firms listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a form of company with a board of auditors. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.

Table 9 presents the results of the difference-in-differences analysis. It presents the mean of the increments for each variable over a given period, and the numbers below represent the number of observations. A significant difference is observed at the 10% level in the increment from period t-2 to period t in Q; however, all the other values are insignificant. This tendency is true for Female1 and Male1. Thus, tokenism does not affect the performance of female outside directors or the performance of male outside directors, even when the number of directors has increased. This may be because many Japanese firms focus on meeting the Code's requirements. However, the change does not appear to impact corporate performance, unless it is significant such as in the case of most boards comprising outside directors. Therefore, we find no impact of tokenism on performance.

7. Conclusion

We take the early stage of the introduction of the Code as a quasi-natural experiment to check whether women's participation in management is observed in Japan, similar to other countries. In the early stages following the Code's introduction, companies focused on appointing two outside directors to meet the requirements. Companies with a small ratio and few female outside directors tried to comply to the requirement. However, there is no indication that they actively attempted to employ female outside directors. To meet the Code's requirement, one male outside director was first appointed, followed by a woman outside director as tokenism. Unlike in other countries, there is no evidence of external pressure. Although there is a tendency to appoint female outside directors when the CEO is female, some results indicate the existence of tokenism. The number of female directors appears to increase as needed while taking care of existing male directors. Furthermore, the departure of female outside directors does not lead to the appointment of new female outside directors. However, we find that the appointment of a female as the second outside director is tokenism when there is one female auditor—a position unique to Japan.

When a busy female outside director is appointed, tokenism does not exist, even if a male is appointed as the first outside director and a female is appointed as the second outside director. Busy female outside directors may be expected to have more experience and skills; therefore, they are less suited for a token appointment. In Japan, where the role of women in corporate management has hardly progressed, the Code's introduction has undeniably promoted the appointment of female outside directors. However, this does not seem to have had a significant economic impact. Additionally, tokenism is found in companies that meet the external criteria of the Code requirements; although, it is not observed when female outside directors have experience and skills.

This study has practical implications. When appointing female outside directors in the future, organizing the social infrastructure (e.g., introducing a quarter system²¹ and skill matrix) and establishing a transparent appointment system (e.g., clarifying the appointment process through a nominating committee) is necessary. Furthermore, companies must make efforts to realize "task-based" diversity,²² in which appointments are made based on abilities, skills, and experience. If socioeconomic conditions help narrow the gender gap, then the number of female outside director candidates increases and companies can select diverse directors from a larger pool of high-quality human resources regardless of gender. Moreover, as argued in Rosenblum (2019), develop legislation

²¹ In the U.S., there has been mixed debate over rulemaking to enhance board member diversity; Nasdaq's new rule to enhance board member diversity in 2021 was welcomed by the industry and practice community but has become a litigious issue with some taking a conservative view. In addition, a California law requiring publicly traded companies to increase board diversity in 2022 has been ruled unconstitutional (Nikkei 2022/9/2).

²² Joshi and Roh (2009) and Horwitz and Horwitz (2007) and is said to be "demographic" diversity when classified by gender, etc., and "task" diversity when classified by ability, skills, and experience.

is sufficient and it is more important to coordinate public efforts with private efforts, that is, it is essential that legislation and institutional investors work together.

However, this study has several limitations. This study focuses on the early days of the Code's introduction. The Code has since required appointing female directors. Recently, TSE required efforts to achieve the proportion of female directors/auditors and senior executives to be more than 30%. There is also a strong request from the so-called "Big 3" foreign investors, as shown by Gormley et al. (2023). Hence, extending the analysis period could yield more useful conclusions; this can be addressed in the future. Furthermore, unlike Coles et al. (2008), this study does not determine the optimal size of the board of directors. Increasing the board of directors may involve appointing or replacing incumbent male directors. As noted by Knippen et al. (2019), considering the motivations and determinants of changes in board size is an interesting topic that can be explored in future research.

Acknowledgements

The author would like to thank Takuji Saito (Keio University), Toshiki Honda (Hitotsubashi University), and the participants of the 2023 Japan Society of Monetary Economics, 2022 Fall Conference of the Nippon Finance Association, and 2022 Hitotsubashi University Academic Conference, and Aoyama Gakuin Finance Seminar for their helpful comments on this paper.

Funding

This work was supported by the JSPS Grants-in-Aid for Scientific Research [grant numbers 20K13529, 20K01750, and 21H04394].

References

- Adams, R., D. Ferreira, 2009. Women in the boardroom and their impact on governance and performance. Journal of Financial Economics 94(2), 291-319. https://doi.org/10.1016/j.jfineco.2008.10.007.
- Aoki, M., Gregory, J., Miyajima, H., 2007. Corporate governance in Japan: institutional change and organizational diversity. Social Science Japan Journal 11(2), 339-343. https://doi.org/10.1093/ssjj/jyn041.
- Bodenhausen, G., Kang, S.K., Peery, D., 2012. Social categorization and the perception of social groups. The SAGE Handbook of Social Cognition: Sage Publications Ltd.
- Campbell, J., 2007. Why would corporations behave in socially responsible ways? an institutional theory of corporate social responsibility. Academy of Management Review 32(3), 946-967. https://doi.org/10.5465/amr.2007.25275684.
- Chizema, A., Kamuriwo, D., Shinozawa, Y., 2015. Women on corporate boards around the world: triggers and barriers. Leadership Quarterly 26 1051-1065. https://doi.org/10.1016/j.leaqua.2015.07.005.
- Coles, J.L., Daniel, N.D., Naveen, L., 2008. Boards: does one size fit all? Journal of Financial Economics 87(2), 329- 356.. https://doi.org/10.1016/j.jfineco.2006.08.008.
- Dechow, P.M., Sloan, R.G., Sweeney, A.P., 1995. Detecting earnings management. The Accounting Review 70(2), 193-225.
- Farrell, K.A., Hersch, P.L., 2005. Additions to corporate boards: the effect of gender. Journal of Corporate Finance 11, 85-106. https://doi.org/10.1016/j.jcorpfin.2003.12.001.
- Field, L.C., Southern, M.E., Yore, A.S., 2020. At the table but can't break through the glass ceiling: board leadership positions elude diverse directors. Journal of Financial Economics 137(3), 787-814. https://doi.org/10.1016/j.jfineco.2020.04.001.
- Gillan, S.L., 2006. Recent developments in corporate governance: an overview. Journal of Corporate Finance 12, 381-402. https://doi.org/10.1016/j.jcorpfin.2005.11.002.
- Gormley, T.A., Gupta, V.K., Matsa, D.A., Mortal, S.C., Yang, L., 2023. The Big Three and board gender diversity: The effectiveness of shareholder voice, Journal of Financial Economics 149(2), 323-348, https://doi.org/10.1016/j.jfineco.2023.04.001
- Green, C.P., Homroy, S., 2018. Female directors, board committees and firm performance. European Economic Review 102, 19-38. https://doi.org/10.1016/j.euroecorev.2017.12.003.
- Gregory-Smith, I., Main, B.G., O'Reilly, C.A., 2014. Appointments, pay and performance in UK boardrooms by gender. The Economic Journal 124 (574), 109-128. https://doi.org/10.1111/ecoj.12102.

- Guldiken, O., Mallon, M.R., Fainshmidt, S., Judge, W.Q., Clark, C.E., 2019. Beyond tokenism: how strategic leaders influence more meaningful gender diversity on boards of directors. Strategic Management Journal 40(12), 2024-2046. https://doi.org/10.1002/smj.3049.
- Harjoto, M.A., Laksmana, I., Yang, Y.-W., 2018. Board diversity and corporate investment oversight. Journal of Business Research 90 (September 2018), 40-47. https://doi.org/10.1016/j.jbusres.2018.04.033.
- Hart, P., 1991. Irving L. Janis' victims of groupthink. Political Psychology 12(2), 247-278. https://doi.org/10.2307/3791464.
- Hillman, A.J., Cannella, A.A., Harris, I.C., 2002. Women and racial minorities in the boardroom: how do directors differ? Journal of Management 28(6) 747-763. https://doi.org/10.1016/S0149-2063(02)00192-7.
- Horwitz, S.K., Horwitz, I.B., 2007. The effects of team diversity on team outcomes: a metaanalytic review of team demography. Journal of Management 33(6), 987-1015. https://doi.org/10.1177/0149206307308587.
- Hutchinson, M., Mack, J., Plastow, K., 2015. Who selects the 'right' directors? An examination of the association between board selection, gender diversity and outcomes. Accounting and Finance 55(4), 1071-1103. https://doi.org/10.1111/acfi.12082.
- Ito, T., T. Hoshi, 2020. The Japanese Economy, Second Edition. The MIT Press.
- Joecks, J., Pull, K., Vetter, K., 2013. Gender diversity in the boardroom and firm performance: what exactly constitutes a 'critical mass? ' Journal of Business Ethics 118(1), 61-72. https://doi.org/10.1007/s10551-012-1553-6.
- Joshi, A., Roh, H., 2009. The role of context in work team diversity research: a meta-analytic review. Academy of Management Journal 52(3), 599-627. https://doi.org/10.5465/amj.2009.41331491.
- Kanter, R. 1977. Men and Women of the Corporation. Basic Books.
- Kirsch, A., Wrohlich, K., 2020. More women on supervisory boards: increasing indications that the effect of the gender quota extends to executive boards. DIW Weekly Report, DIW Berlin 10(4/5), 44-49.
- Kim, D., Starks, L.T., 2016. Gender diversity on corporate boards: do women contribute unique skills? Discovery Service for UNISA. American Economic Review: Papers & Proceedings 106(5), 267-271. https://doi.org/10.1257/aer.p20161032.
- Knippen , J., Shen, W., Zhu, Q., 2019. Limited progress? the effect of external pressure for board gender diversity on the increase of female directors. Strategic Management Journal 40(7), 1123-1150. https://doi.org/10.1002/smj.3014.
- Konrad, A., Kramer, V., Erkut, S., 2008. Critical mass: the impact of three or more women on corporate boards. Organizational Dynamics 37(2), 145-164.

- Kubo, K., Nguyen, T.T.P., 2021. Female CEOs on Japanese corporate boards and firm performance. Journal of the Japanese and International Economies 62, 101163. https://doi.org/10.1016/j.jjie.2021.101163.
- Matsui, R., 2020. What is groupthink? the possibility of groupthink in complex organizations. Journal of the Atomic Energy Society of Japan 62(5), 272-276.
- Mun, E., Jung, J., 2018. Change above the glass ceiling: corporate social responsibility and gender diversity in Japanese firms. Administrative Science Quarterly 63(2), 409-440. https://doi.org/10.1177/0001839217712920.
- Nekhili, M., Gatfaoui, H., 2013. Are demographic attributes and firm characteristics drivers of gender diversity? investigating women's positions on French boards of directors. Journal of Business Ethics 118(2), 227-249. https://doi.org/10.1007/s10551-012-1576-z.
- Nguyen, T.T.P., Thai, H.M., 2022. Effects of female directors on gender diversity at lower organization levels and CSR performance: evidence in Japan. Global Finance Journal 53, 100749. https://doi.org/10.1016/j.gfj.2022.100749.
- Post, C., Byron, K., 2015. Women on boards and firm financial performance: a meta-analysis. Academy of Management Journal 58(5), 1546-1571. https://doi.org/10.5465/amj.2013.0319.
- Raddant, M., Takahashi, H., 2022. Interdependencies of female board member appointments. International Review of Financial Analysis 81, 102080. https://doi.org/10.1016/j.irfa.2022.102080.
- Rosenblum, D., 2019 California Dreaming, Boston University Law Review, Vol. 99, 1435 (2019)
- Shirai, S., 2022. Global trends in carbon neutrality-policy, money and civil society. Bunshindo (in Japanese).
- Sila, V., Gonzalez, A., Hagendorff, J., 2016. Women on board: does boardroom gender diversity really affect firm risk? Journal of Corporate Finance 36, 26-53. https://doi.org/10.1016/j.jcorpfin.2015.10.003.
- Süssmuth-Dyckerhoff, C., Wang, J., Chen, J., 2012. Women Matter: An Asian Perspective. McKinsey & Company. https://www.mckinsey.com/~/media/McKinsey/dotcom/client_service/Organization/PDFs/

Women Matter Asia.ashx

- Tanaka, T., 2019. Gender diversity on Japanese corporate boards. Journal of the Japanese and International Economies 51, 19-31. https://doi.org/10.1016/j.jjie.2018.08.003.
- Zhu, D.H., Shen, W., Hillman, A.J., 2014. Recategorization into the in-group: the appointment of demographically different new directors and their subsequent positions on corporate

boards. Administrative Science Quarterly 59(2), 240-70. https://doi.org/10.1177/0001839214530951.

			Fem	ale1			Ma	le1	
		Treat	Control	Difference	t-stat	Treat	Control	Difference	t-stat
ROA	Full sample	11.2%	9.8%	1.3%	1.97 **	9.5%	9.9%	-0.4%	-1.22
		88	4,125			447	3,766		
	PSM	13.0%	10.6%	2.4%	0.88	9.6%	9.8%	-0.2%	-0.33
		35	35			227	227		
Q	Full sample	1.460	1.392	0.068	0.54	1.259	1.410	-0.151	-2.58 ***
		88	4,125			447	3,766		
	PSM	1.662	2.053	-0.390	-0.81	1.287	1.394	-0.106	-1.14
		35	35			227	227		
ABACC	Full sample	0.025	0.026	-0.001	-0.37	0.031	0.026	0.005	3.24 ***
		79	3,708			406	3,381		
	PSM	0.027	0.033	-0.005	-0.78	0.032	0.027	0.005	1.49
		32	32			208	205		
Audit Fee	Full sample	120.0	90.1	29.9	1.28	64.6	93.8	-29.2	-2.64 ***
		87	3,938			427	3,598		
	PSM	123.1	63.9	59.3	1.27	61.9	79.8	-17.9	-1.34
		35	35			218	206		

Appendix 1 Univariate comparison

Table A-1 shows univariate comparisons of the outcome variables used in the difference-in-differences analysis. For each of the outcome variables, the mean before PSM (Full sample) and after PSM (PSM) are shown. The number at the bottom of the mean represents the number of observations. The sample analyzed comprises general business firms listed on the First Section of the Tokyo Stock Exchange in the fiscal years 2014-2017 that have a form of company with a board of auditors. *, **, and *** represent significance at the 10%, 5%, and 1% levels, respectively.